UCLA UCLA Previously Published Works

Title

Bayesian phylogeography of influenza A/H3N2 for the 2014-15 season in the United States using three frameworks of ancestral state reconstruction

Permalink

https://escholarship.org/uc/item/8mq251j2

Journal PLOS Computational Biology, 13(2)

ISSN

1553-734X

Authors

Magee, Daniel Suchard, Marc A Scotch, Matthew

Publication Date

2017

DOI

10.1371/journal.pcbi.1005389

Copyright Information

This work is made available under the terms of a Creative Commons Attribution License, available at https://creativecommons.org/licenses/by/4.0/

Peer reviewed



Citation: Magee D, Suchard MA, Scotch M (2017) Bayesian phylogeography of influenza A/H3N2 for the 2014-15 season in the United States using three frameworks of ancestral state reconstruction. PLoS Comput Biol 13(2): e1005389. doi:10.1371/ journal.pcbi.1005389

Editor: Katia Koelle, Duke University, UNITED STATES

Received: July 6, 2016

Accepted: January 27, 2017

Published: February 7, 2017

Copyright: © 2017 Magee et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files. We have provided all of our XML files, MCC trees, and sequence alignments in a public repository (https://figshare.com/projects/Magee-Flu-PLoS/16638).

Funding: Research reported in this publication was supported by the National Library of Medicine of the National Institutes of Health under Award Number R01LM012080 (to MS). The content is solely the responsibility of the authors and does

RESEARCH ARTICLE

Bayesian phylogeography of influenza A/H3N2 for the 2014-15 season in the United States using three frameworks of ancestral state reconstruction

Daniel Magee^{1,2}, Marc A. Suchard^{3,4}, Matthew Scotch^{1,2}*

Department of Biomedical Informatics, Arizona State University, Tempe, Arizona, United States of America,
Biodesign Center for Environmental Security, Arizona State University, Tempe, Arizona, United States of America,
Departments of Biomathematics and Human Genetics, David Geffen School of Medicine,
University of California, Los Angeles, California, United States of America,
Department of Biostatistics,
School of Public Health, University of California, Los Angeles, California, Los Angeles, California, United States of America

* matthew.scotch@asu.edu

Abstract

Ancestral state reconstructions in Bayesian phylogeography of virus pandemics have been improved by utilizing a Bayesian stochastic search variable selection (BSSVS) framework. Recently, this framework has been extended to model the transition rate matrix between discrete states as a generalized linear model (GLM) of genetic, geographic, demographic, and environmental predictors of interest to the virus and incorporating BSSVS to estimate the posterior inclusion probabilities of each predictor. Although the latter appears to enhance the biological validity of ancestral state reconstruction, there has yet to be a comparison of phylogenies created by the two methods. In this paper, we compare these two methods, while also using a primitive method without BSSVS, and highlight the differences in phylogenies created by each. We test six coalescent priors and six random sequence samples of H3N2 influenza during the 2014–15 flu season in the U.S. We show that the GLMs yield significantly greater root state posterior probabilities than the two alternative methods under five of the six priors, and significantly greater Kullback-Leibler divergence values than the two alternative methods under all priors. Furthermore, the GLMs strongly implicate temperature and precipitation as driving forces of this flu season and nearly unanimously identified a single root state, which exhibits the most tropical climate during a typical flu season in the U. S. The GLM, however, appears to be highly susceptible to sampling bias compared with the other methods, which casts doubt on whether its reconstructions should be favored over those created by alternate methods. We report that a BSSVS approach with a Poisson prior demonstrates less bias toward sample size under certain conditions than the GLMs or primitive models, and believe that the connection between reconstruction method and sampling bias warrants further investigation.

not necessarily represent the official views of the National Institutes of Health. We would also like to thank the ARCS Foundation, Phoenix Chapter for their generous support (to DM). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

For the better part of the last decade, epidemiological researchers have employed a Bayesian framework to reconstruct phylogenetic trees and determine the spatiotemporal relationships between clades of viruses. Recently, an extension of this framework has enabled direct assessment of how various demographic, geographic, genetic, and environmental variables play a role in these relationships, but there has yet to be a comparison between the former and the latter. Here, we aim to assess the differences between the two reconstruction techniques, as well as an additional primitive method, using the 2014–15 influenza season in the U.S. as a case study under a variety of population growth scenarios. We highlight how the new method demonstrates significant increases in commonly-reported trends in phylogenies and that the method identifies climate predictors that appear to be consistent with known trends in seasonal trends in influenza. However, we found that this method appears to be the most heavily influenced by the locations at which the viruses were obtained. Our work offers valuable insight for researchers wishing to study the evolutionary history of viruses and also may prove useful in determining the correct method to choose for a given application of virus phylogeography.

Introduction

Bayesian phylogeography has emerged as a powerful approach to analyzing virus spread. It utilizes sequence data to perform ancestral reconstruction and estimate the most likely lineages of the viruses in rooted, time-measured phylogenies [1] using nucleotide substitution models, molecular clocks, and coalescent priors under a probabilistic Bayesian framework known as Bayesian stochastic search variable selection (BSSVS) [1-3]. This framework has improved ancestral state reconstruction and has recently been used to analyze human and animal influenza viruses both globally [4-5] and nationally [6-7]. By identifying the relationship between geospatial origins and genetic lineages, much can be learned about the complex process in which these viruses spread. Phylodynamic analyses that aim to combine immunological, epidemiological, and evolutionary biology techniques [8] also enhance our understanding of virus transmission dynamics and their relationship to a phylogeny. These studies have unveiled novel properties of several influenza viruses, including pdm09 [9], H3N2 [10] and highly pathogenic avian influenza H5N1 [11]. Building upon the benefits of a BSSVS framework, recent work by Lemey et al. [12] utilized a phylogeographic generalized linear model (GLM) approach to identify environmental, genetic, demographic, and geographic predictors that contributed to the global spread of H3N2 influenza viruses. In the GLM, the BSSVS on the discrete location variable is instead used to estimate the posterior inclusion probability of potential predictors in a log-linear combination to model the transition rate matrix. Similarly, studies have followed this approach to uncover the predictors associated with the spread of H5N1 in Egypt [13] and for HIV in Brazil [14]. Such studies have demonstrated the utility of combining genetic and geospatial inferences from phylogeography with surveillance data in epidemiological studies like Yang et al. [15]. These analyses may enable actionable solutions for public health officials once consistent identification of contributing predictors is achieved.

Although the GLM appears to show promise with its simultaneous ability to perform ancestral state reconstruction and also assess the contribution of predictor variables of interest, there has yet to be an assessment of how a standard BSSVS approach and a GLM approach differ in reconstructing a phylogeny. Specifically, no study has yet compared root state probabilities in a phylogeny constructed via BSSVS to the same probabilities using the GLM approach. Such information may inform researchers of differences in phylogeographic trends that may be experienced by choosing one framework over the other. In this work we analyze the 2014–15 H3N2 flu season within the U.S. by performing ancestral state reconstruction of a discrete location variable via the following three frameworks: an asymmetric substitution model without BSSVS (-BSSVS), an asymmetric substitution model with BSSVS (+BSSVS) [1], and a GLM [12]. For the BSSVS framework, we analyze separate versions that place both a Poisson distribution (+BSSVS(P)) and a uniform distribution (+BSSVS(U)) on the number of rate parameters that achieve a point-mass on 1.0 in order to determine the influence of location priors. For the GLM framework, we analyze separate versions that include and do not include sample size predictors, which we denote as GLM(+SS) and GLM(-SS), respectively, in order to directly quantify the effect of sampling bias on GLM-constructed rate matrices and potential suppression of the signal of other predictors. This brings us to a total of five methods that encompass the three frameworks. We refer readers to Materials and methods for full details on the methods. These selections allow us to empirically evaluate differences in phylogenies obtained via each method and to determine whether one framework provides more accurate posterior estimates given a fixed set of data. We demonstrate these trends using multiple random samples from a large collection of flu sequences to show reproducibility as well as analyze several coalescent tree priors to show consistency among the reconstruction methods across varying parameters. Finally, we show that support for GLM predictors can change given the tree priors and sequence sets, but that trends among specific predictors will emerge to allow accurate determination of their impact on viral diffusion.

Results

In Fig 1A, we show mean log marginal likelihood estimates among the six samples obtained by path sampling (PS) and stepping stone sampling (SSS) for each prior and reconstruction method. For PS, the two best-performing mean methods are the GLM(+SS) and GLM(-SS), respectively, under each prior. The mean +BSSVS(U) outperforms the mean +BSSVS(P) under each prior as well, although the mean -BSSVS exceeds both under the constant and exponential priors. For SSS, the log marginal likelihood increases in a near-linear manner for the +BSSVS(P), +BSSVS(U), GLM(-SS), and GLM(+SS) methods. The -BSSVS method, however, finds the largest posterior support under the constant, expansion, exponential, logistic, and Skyline priors.

In <u>S1 Fig</u>, we present log marginal likelihood estimates for each individual model. From <u>S1</u> Fig, we show that each GLM(+SS) and GLM(-SS) unanimously finds more posterior support than their corresponding +BSSVS(P) for both PS and SSS. The +BSSVS(P) method demonstrates consistently poor performance, as its posterior estimates are the worst of the five methods in 25 of 36 PS analyses and 32 of 36 PS analyses (79% overall) across all priors, while no GLM(+SS) or GLM(-SS) yields the lowest posterior estimate of model support among the three methods for either PS or SSS under any prior, although no pairwise t-test shows a significant difference.

Each of the 180 models show statistically significant differences between the null and observed means for the association index (S2 Fig). These data suggest stronger support for the phylogeny-trait association [16] and, as all p < 0.01, suggest the evolution of influenza during this flu season was structured by geography. The support of the sampling location-phylogeny associations observed in S2 Fig can be explained, in part, by the amount of genetic diversity observed within and across each region. In Fig 1B we show the average genetic distances between intra-region and inter-region sequences. Here, we calculated the genetic distances



Fig 1. Model comparison statistics and location-specific genetic diversity. (A) Model comparisons obtained via path sampling (PS) and stepping stone sampling (SSS) for the six coalescent priors and five methods. (B) Average genetic distances between all pairwise intra-region and inter-region sequences for the six samples, expressed as a percent, with 95% confidence intervals shown as error bars.

among all $\binom{285}{2}$ pairwise sequences and present the mean distance of sequences sampled in

the same region (e.g. Region 1-Region 1) to those sampled in different regions (e.g. Region 1-Region 2). From Fig 1B, the pairwise intra-region sequences (n = 4,496 per sample) have a lesser amount of genetic diversity than the pairwise inter-region sequences (n = 35,974 per sample) in each our six sequence sets. A two-tailed t-test shows p < 0.01 for each sample, indicating that sequences from within the same region demonstrate significantly lower amounts of genetic diversity than those from external regions. The average intra- and inter-region distances in the full set of 1,163 sequences are 0.872% (95% CI = [0.867, 0.878]), and 0.929% (95% CI = [0.926, 0.932]), respectively (p < 0.0001). These data demonstrate that our method of downsampling maintained representative levels of genetic diversity across the six samples.

In Fig 2, we show four root state metrics obtained from the maximum clade credibility (MCC) trees of each of the 180 models. In Fig 2A, we show the mean root state posterior probability (RSPP). Aside from the constant coalescent prior, the mean GLM(–SS) and GLM(+SS) methods consistently show the largest mean RSPP of the five methods. The mean GLM(–SS) finds significantly greater RSPPs under each coalescent prior than the mean -BSSVS (p < 0.03 for each coalescent prior) and significantly greater RSPPs than both the mean +BSSVS(P) and +BSSVS(U) for the expansion and exponential coalescent priors. Similarly, the GLM(+SS) shows a mean RSPP significantly greater than the -BSSVS and +BSSVS(U) methods for all coalescent priors except constant, and significantly greater RSPP than the +BSSVS(P) for the constant, expansion, Skygrid, and Skyline coalescent priors. Across all coalescent priors, the mean RSPP for the -BSSVS, +BSSVS(P), +BSSVS(U), GLM(–SS), and GLM(+SS) methods are 0.48, 0.56, 0.49, 0.81, and 0.74 respectively. These differences per method could be influenced



Fig 2. Mean posterior metrics of the MCC phylogenies. Values represent the mean indicated statistic from the six samples under each coalescent prior and method with error bars representing the standard error. (A) Root state posterior probability. (B) Pearson's correlation coefficient for the number of sequences per discrete state and the root state posterior probability for each discrete state in each model. (C) Kullback-Leibler divergence calculated assuming a uniform prior probability per discrete state. (D) Kullback-Leibler divergence calculated assuming a prior probability proportional to the number of sequences per discrete state.

by the sample size per discrete state, so we show the Pearson's r correlation coefficient between the sample size at each discrete state and its corresponding posterior probability at the root in Fig 2B. Here we observe that the +BSSVS(P) shows a correlation coefficient less than 0.4 for the constant, expansion, Skygrid, and Skyline coalescent priors but for the exponential and logistic coalescent priors the coefficient is nearly doubled. Meanwhile, the +BSSVS(U), -BSSVS, GLM(–SS), and GLM(+SS) methods are generally consistent under all priors. The mean +BSSVS(P) shows significantly less correlation than each of the other four methods for the constant, expansion, and Skyline coalescent priors (p < 0.02 for each) while the +BSSVS (U), -BSSVS, and GLM methods do not show any significant differences under any coalescent prior.

Fig 2C and 2D show the Kullback-Leibler (KL) divergence between the prior and posterior probabilities at the root states calculated using two different prior assumptions (see Materials and methods for details). KL values indicate the extent to which a model is able to generate different posterior probabilities at the root state from the prior probabilities at the root state. That is, high KL values indicate strong divergence from the prior probabilities and, thus, strong posterior information gain, while low KL values indicate the opposite. From Fig 2C and 2D, the mean GLM(–SS) and GLM(+SS) KL divergences demonstrate a marked increase over the -BSSVS, +BSSVS(P), and +BSSVS(U) methods under the expansion, exponential, logistic, Skygrid, and Skyline coalescent priors (p < 0.02 for all two-tailed t-tests. Under the constant coalescent prior, both the mean GLM(–SS) and GLM(+SS) KL divergences exceed the mean KL under both assumptions of the -BSSVS, +BSSVS(P), and +BSSVS(U) methods, but none of

Method -BSSVS +BSSVS(P)	State									
	1	2	3	4	5	6	7	8	9	10
-BSSVS	_	_	-	23	11	2	_	-	_	-
+BSSVS(P)	_	2	1	10	6	16	-	1	_	-
+BSSVS(U)	_	_	-	17	10	9	-	-	_	_
GLM(-SS)	_	_	-	35	-	_	_	1	_	-
GLM(+SS)	_	_	_	33	3	_	_	_	_	_

Table 1. Frequencies of the root states identified in the MCC tree under each reconstruction method.

these values are significant. The +BSSVS(P) method, in turn, shows significantly greater KL divergences under both assumptions than the -BSSVS method under all coalescent priors and than the +BSSVS(U) method under the constant, exponential, and logistic coalescent priors. We show data for each of the four metrics in Fig 2 by individual model in S3 and S4 Figs.

We summarize the identified root states of the four methods in Table 1. Here, we can see that the -BSSVS method identified three different regions, with the majority occurring in Region 4, while Region 5 is identified in over 30% of -BSSVS models. The +BSSVS(P) method identified six different regions as the root state, with Regions 6 and 4 representing the most frequently-identified. The +BSSVS(U) method identified Region 4 in nearly half of the models while Regions 5 and 6 account for the remainder of models. Comparatively, 35 of the 36 GLM (-SS) runs identified Region 4 as the root state, with the lone exception being Sample 2 using the Skygrid coalescent prior, which identified Region 8. For the GLM(+SS) analyses, Region 4 is identified as the root state in 33 of 36 models while Region 5 accounts for the remaining three. The root heights and corresponding Bayesian credible intervals are similar between the three methods for each sample and each coalescent prior (S5 Fig).

As influenza viruses rarely persist for more than one season, except in tropical areas [17, 18], we obtained the geographic distribution of the number of internal nodes with a height of at least one year (NH1s) from the MCC tree of each model and show these data in Fig 3A. From Fig 3A, we can see that the -BSSVS method indicates that Region 4 contains the highest volume of NH1s under each prior, while Region 5 contains the second-largest volume of NH1s. The +BSSVS(P) method shows Region 4 containing the most NH1s for the exponential, logistic, Skyline, and Skygrid coalescent priors, with Region 6 accounting for the next largest volume in the latter three priors. Under the constant coalescent prior, a nearly equal amount of NH1s are observed in Regions 4, 6, and 8, while the expansion prior shows Region 5 containing the largest number of NH1s. For the +BSSVS(U) method, the NH1s are most commonly observed in Region 4 under each coalescent prior, with Regions 5 and 6 primarily accounting for the remaining nodes. The frequency of NH1s in Region 8 are low under this method, but do occur under the constant, expansion, and Skygrid coalescent priors. Finally, the NH1s are largely concentrated in Region 4 for both the GLM(–SS) and GLM(+SS) methods under each coalescent prior.

The frequent identification of Region 4 as the root state (Table 1) and location of NH1 events (Fig 3A) indicates that there is likely at least one local variable playing a role in the tree topologies. Given this, from Fig 3B we note that Region 4 exhibits both the highest expected temperature and precipitation during a typical flu season as we compare the posterior support of all predictors for both the GLM(–SS) and GLM(+SS) methods in Fig 4.

From Fig 4, we can see that sample size at the region of origin (SS(O)) is strongly supported for the GLM(+SS) runs with Bayes factor (BF) > 69 for each coalescent prior and with each corresponding mean regression coefficient greater than 1.33. The predictor with the second







largest support for inclusion in the GLM(+SS) runs is temperature at the region of origin (BF > 5 and regression coefficient > 0.75 for each prior except constant size), followed by gly-coprotein at the region of origin (3.0 < BF < 4.5 for the expansion, exponential, Skyline, and Skygrid coalescent priors) although the respective mean regression coefficients for



Fig 4. Mean posterior estimates of supported predictors. We show the inclusion probabilities and regression coefficients for all predictors for both the GLM(–SS) and GLM(+SS) analyses. Point estimates represent the mean of each statistic across the six models for each prior, with error bars representing the standard error of these estimates. Predictor abbreviations are: air travel (AT), glycoprotein content (GP), median age (MA), precipitation (PC), population density (PD), sample size (SS), temperature (TP) and vaccination rate (VR).

glycoprotein remain near zero. For the GLM(-SS) runs, temperature at the region of origin yields the largest mean posterior inclusion probability across all coalescent priors (BF > 20 for each prior, BF > 400 for the expansion, exponential, logistic, and Skyline priors) followed by precipitation at the region of origin (5.0 < BF < 8.5 for all priors). Mean posterior estimates of the corresponding regression coefficients and their standard errors, shown as $E(\beta|\delta = 1)$, indicate strictly positive values for these two predictors in the GLM(-SS) runs, although the 95% highest posterior density (HPD) of the regression coefficient for precipitation at the region of origin spans zero for each model (S6 Fig). If the entire HPD lies on the positive side of zero, this suggests that the predictor is driving the diffusion of the virus. Conversely, if the entire HPD lies on the negative side of zero, this suggests that the predictor is rather preventing the diffusion. Thus, we show the proportion of GLMs in which the absolute value of the HPD is positive in Table 2. The 95% HPDs of temperature at the region of origin are strictly positive in 26 of the 36 GLM(-SS) runs and span zero in the remaining ten. The glycoprotein predictor at the region of origin finds the highest mean support for the constant prior (BF = 1.1), which is a sharp turn from the GLM(+SS) runs. See Materials and methods for more information on metrics of support and interpretations of our predictors. We show the posterior regression

Method	Criterion	Predictor at the Region of Origin								
		AT	GP	MA	PC	PD	SS	TP	VR	
GLM(-SS)	BF ≥ 3	-	3%	25%	36%	3%	NA	94%	19%	
GLM(+SS)	$BF \ge 3$	_	17%	-	3%	-	97%	36%	3%	
GLM(-SS)	95% HPD (β) > 0	-	-	-	-	-	NA	72%	-	
GLM(+SS)	95% HPD (β) > 0	-	3%	-	-	-	61%	8%	-	

Table 2. Frequency of GLM predictor support.

Values represent the percentage of models that show BF support for a predictor and the percentage of 95% HPD intervals of the regression coefficient that do not span zero. Predictor abbreviations are: air travel (AT), glycoprotein content (GP), median age (MA), precipitation (PC), population density (PD), sample size (SS), temperature (TP) and vaccination rate (VR).

doi:10.1371/journal.pcbi.1005389.t002

coefficients and inclusion probabilities of every predictor from each of the 36 GLM(–SS) runs in <u>S6</u> and <u>S7</u> Figs, respectively, and corresponding data for the 36 GLM(+SS) runs in <u>S8</u> and <u>S9</u> Figs, respectively.

Discussion

In this paper, we compared three ancestral state reconstruction frameworks and five total methods using six randomly-drawn sequence samples and six coalescent priors for a total of 180 models while fixing the nucleotide substitution process for each. We compared each of our analyses with established model selection techniques [19, 20] and compared features of each model's MCC tree to identify posterior statistical support and discrepancies in the phylogeographic reconstructions. Regarding model selection, we found that PS shows the most posterior support for either the GLM(–SS) or GLM(+SS) in 34 of 36 runs (with one -BSSVS and one +BSSVS(U) accounting for the remaining two), while SSS shows the most support for 29 of 36 –BSSVS models, five GLM(+SS), one GLM(–SS), and one +BSSVS(U). Each GLM(–SS) and GLM(+SS) outperformed its corresponding +BSSVS(P) under both PS and SSS. Both statistics agree that +BSSVS(P) models offered the poorest posterior support, as 72% of PS analyses and 89% of SSS analyses (81% combined) show the +BSSVS(P) model as the least-supported among the five frameworks (Fig 1A and S1 Fig), although we note that no framework shows significantly more support than any other framework for PS or SSS via t-tests.

Although the -BSSVS method is highly supported under SSS, the method fails to find strong support regarding both RSPP and KL divergence (Fig 2C, 2D and S4 Fig). The RSPPs using the -BSSVS method are significantly lower than those obtained via the GLM(-SS) method (p = 0.03 for the constant coalescent prior, p < 0.001 for the expansion, exponential, logistic, Skygrid, and Skyline coalescent priors), while the GLM(-SS) also show a significant increase for KL divergence for both the uniform and sample size assumptions over the -BSSVS models under each coalescent prior except for constant size. Similarly, the GLM(+SS) method shows significantly greater RSPPs and both KL divergences than the -BSSVS models (p < 0.03for all coalescent priors except constant). Meanwhile, the +BSSVS(P) method finds significantly greater RSPPs than the -BSSVS method under only the constant coalescent prior (p < 0.001) and significantly greater KL divergences over the -BSSVS method under each coalescent prior, each with p < 0.03. The +BSSVS(P) method also found significantly greater KL divergences for the constant, exponential, and logistic coalescent priors. The +BSSVS(U) method only found significantly greater support over the -BSSVS method via KL with the sample size assumption for the expansion coalescent prior. While these results show that the -BSSVS method finds poor statistical support at the identified root state, we also found that

both the GLM(–SS) and GLM(+SS) methods in turn significantly outperformed both the +BSSVS(P) and +BSSVS(U) models both for both KL divergences under five of the six coalescent priors (excluding constant). The GLM(–SS) runs also found significantly greater RSPPs than the +BSSVS(P) and +BSSVS(U) under each coalescent prior except constant, while the GLM(+SS) runs found significantly greater RSPPs than the +BSSVS(P) and +BSSVS(U) methods for the expansion, Skygrid, and Skyline priors.

The association index of each model obtained via BaTS (S2 Fig) demonstrate a strong association between sampling location and the phylogeny for each of the 180 models, which suggests that the diffusion was spatially-structured. Some of the phylogeny-location association can be attributed to the smaller amount of genetic diversity in sequences from the same region (Fig 1B), however the statistical significance of the intra- and inter-region genetic distances could not fully account for the differences in RSPP and KL divergence, regardless of the coalescent prior. Furthermore, Region 4 was the most frequently-identified root state for the -BSSVS, +BSSVS(U), GLM(-SS), and GLM(+SS) methods, the second most frequently identified root state for +BSSVS(P) method (Table 1), and was also the location of the most NH1s (Fig 3A). These NH1s are biologically important for seasonal influenza, as these viruses typically experience bottlenecking at this height as part of a sink-source ecological dynamic [17, 21, 22]. As Region 4 experiences the highest temperature and most precipitation during flu season, at 6.9°C warmer and 10.3 cm wetter, respectively, than the remaining nine regions (Fig 3B) we describe it as the most "tropical" in the U.S. during a typical flu season. This provides a well-supported explanation for the observed trends in Region 4, especially under both GLM methods. As the data for the GLM(-SS) and GLM(+SS) runs indicate strong support for temperature at the region of origin (Fig 4), our results would suggest that Region 4 is the most likely origin of each of the six samples using those two methods.

This conclusion, however, is hindered by the strong sampling bias exhibited by the GLM(-SS), and GLM(+SS) methods. These two methods (as well as the -BSSVS and +BSSVS(U)) demonstrate consistently strong, positive Pearson's r correlation coefficients between the root state posterior probability and sample size at each discrete state, regardless of coalescent prior (Fig 2B and S3B Fig). Furthermore, the inclusion of the sample size predictors in the GLM (+SS) runs shows that sample size at the region of origin is strongly influencing its posterior estimates, with 35 of 36 runs showing BF > 3 and 22 of 36 showing a positive 95% HPD on the regression coefficient (Table 2, S8 and S9 Figs). The mean posterior inclusion probability for the sample size predictor at the region of origin corresponds to BFs of 1317.9, 70.0, 122.9, 102.7, 92.6, and 101.8 for the constant, expansion, exponential, logistic, Skygrid, and Skyline priors, respectively. Given the similarities in RSPP, Pearson's r, and KL data between the GLM (-SS) and GLM(+SS) runs (Fig 2, S3 and S4 Figs), we believe that sample size is influencing the GLM(-SS) runs to a similar degree, although its BF support cannot be measured. Thus, although it would appear that both GLM methods presented in this paper are providing biologically justifiable and statistically supported evidence regarding the diffusion of this influenza virus over our selected time period, the strong sampling biases give us pause. Instead, the significant decrease in Pearson's r for the +BSSVS(P) models from the other four methods under the constant, expansion, and Skyline coalescent priors provide more confidence in those data, despite its poor performance with respect to log marginal likelihoods via PS and SSS (Fig 1A and S1 Fig).

We compared the -BSSVS, +BSSVS(P), +BSSVS(U), GLM(–SS), and GLM(+SS) methods for modeling a single discrete trait, sampling location, which highlighted differences in diffusion of seasonal influenza in the U.S. Our results collectively indicate that the GLMs provide the strongest posterior support for MCC metrics of the three ancestral state reconstruction frameworks used in this study, however the strong sampling bias exhibited by that method marginalizes confidence in their reconstructions. As mentioned, the strong support for sample size is consistent with previous studies that used the phylogeographic GLMs [12, 13]. Air travel was previously shown to be a driver of the global diffusion of H3N2 using a GLM [12], but none of the GLM(–SS) or GLM(+SS) runs showed support for this predictor. However, our study was performed within a single country and aggregated all air travel data from each individual state into a matrix of region-to-region passenger flux, which perhaps limits its contribution to these models. Furthermore, the paper by Lemey *et al.* [12] discretized by "air communities" (p. 2) to better reflect trends in air travel, while we partitioned strictly based on pre-defined, arbitrary geographic regions. We also assumed a single introduction into the U.S. and did not include incoming travel from international flights that could certainly have introduced strains with more genetic diversity than those used in this study.

We recognize several limitations with this study including the omission of international air travel. In addition, our assumption of a single introduction into the U.S. could also have limited inference regarding the contribution of air travel and may explain the lack of BF support for that predictor from both region of origin and destination when a previous study has implicated these data as a driver of the diffusion [12]. Also, the transportation predictor fails to incorporate inter-region travel via ground transportation, which certainly could have implications within a single country. Furthermore, we only analyzed hemagglutinin sequences in this study and did not investigate neuraminidase or any other segments of the influenza genome. We arbitrarily selected 25% of samples from each region for our subsampling in order to better reflect the observed sampling frequencies, but it is possible that larger subsample sizes or an alternative sampling approach could have resulted in stronger or weaker support for the predictors in the GLM as well as the RSPPs via the three reconstruction approaches. However, our use of Pearson's correlation coefficient between sample size and root state posterior probability (Figs 2B and 3B) and comparison of GLMs that include and do not include sample size predictors aim to outline the impact of sampling bias within our dataset. We plan to conduct similar research on additional influenza seasons and using alternative sampling methods in order to further study whether this sampling bias is a systematic function in the GLMs or is limited to the dataset used in this study. Sampling bias is a known issue in phylodynamics [23, 24] and may not be possible to eliminate, although varying approaches may differ in their sensitivity to such biases. Finally, we limited our study to a single influenza season which prevents seasonality comparisons and impacts from local persistence.

Overall, this study aimed to investigate the phylogeography of the H3N2 influenza viruses that circulated in the U.S. during the 2014-15 flu season and to also investigate three established methods of ancestral state reconstruction. While our GLM results provide superior posterior support than either +BSSVS method or the -BSSVS framework, these results appear to be dominated by a strong sampling bias. Although these results are not necessarily incorrect, the investigation of additional frameworks reveals that the +BSSVS(P) is likely the "best" approach for this dataset to minimize such concerns, depending on the selection of coalescent prior, if given the choice among the five presented in our work for this particular virus and time frame. Furthermore, we demonstrate that our approach of subsampling to compare multiple models may not only reflect subtle changes to the phylogeny but also to the contribution of the predictor variables in the GLMs. Although we do not believe that the GLM provides an ideal, unbiased reconstruction framework for our dataset, this type of assessment could be valuable for understanding the true nature of the phylogeny-sampling location association in future work. Such studies may also encourage researchers to utilize the GLM framework as a means of obtaining more information-driven variables into their phylogeographic studies and to unlock the potential for more accurate ancestral state reconstructions to better aid epidemiological and public health efforts.

Materials and methods

Sequence and model setup

Nucleotide sequences. We used the EpiFlu database from the Global Initiative for Sharing All Influenza Data (GISAID) to collect H3N2 hemagglutinin (HA) sequences from the 2014–15 flu season. We obtained our dataset on 2015-10-16 using the following search terms: Host = *Human*, Location = *United States*, Collection Date = 2014-09-29 to 2015-05-17, Submitting Laboratory = [*United States, Atlanta*] *Centers for Disease Control and Prevention*, Required Segments = *HA*, Min Length = 1,659. This search resulted in 1,220 sequences, and we further eliminated sequences from Alaska, Hawaii, and the District of Columbia and those that did not have a specific state listed to obtain a final set of 1,163 sequences. In order to reduce the size of the transition rate matrix, we discretized the states into the ten U.S. Department of Health and Human Services (HHS) regions [25], which we show in Fig 3B.

Ancestral state reconstruction methods. Our phylogeographic assessment assumes that geographic sampling traits follow a continuous-time Markov chain (CTMC) process along the branches of an unknown phylogeny that is informed through sequence data. The models we compare differ in how one parameterizes the infinitesimal rates of the among-location CTMC process. Here, we first parametrized the discrete location trait with a basic asymmetric substitution model (-BSSVS). Next, following Lemey et al. [1], we retained the asymmetric substitution model but specified a truncated Poisson prior on the number of non-zero rates (+BSSVS (P)). Here, 50% of the prior probability lies on the minimal rate configuration (i.e. nine nonzero rates connecting the ten HHS regions). Similarly, we also placed a uniform probability on the location prior in order to test the effects of the selected location prior on the BSSVS procedure +BSSVS(U). We compare the -BSSVS and +BSSVS(P) methods with recent developments in virus phylogeography that have advanced modeling of among-location transition rates as a log-linear GLM of predictors of interest [12]. Here, we followed this framework and parameterized GLMs with seven demographic, environmental, and genetic factors that we take from both region of origin and region of destination for a total of 14 predictors in the GLM(-SS) runs. In the GLM(+SS) runs we also include an additional two sample size predictors for a total of 16 predictors. This approach yields a quantifiable assessment of the inclusion and contribution of each predictor variable to the overall transition rate matrix between our ten locations by estimating posterior probabilities of all 2¹⁴ or 2¹⁶ possible linear models via a BSSVS procedure. We specified a 50% prior probability that no predictor will be included to enable calculation of Bayes factors (BFs) as a metric of support for the inclusion or exclusion of any given predictor. Here, we consider any predictor with BF > 3.0 to be supported for inclusion. For further details on the underlying theory and mathematical definitions of this GLM approach, we refer readers to Lemey et al. [12].

Summary of rate parameters. For both the -BSSVS and +BSSVS frameworks, there are K (K–1) relative rate parameters where K = 10 discrete states for our dataset [1]. For the -BSSVS framework, these rate parameters are each a priori independently gamma distributed with scale and shape parameters of 1.0, and for the +BSSVS framework these rate parameters are each a priori with a mixture of point-mass on 1.0 and on the same gamma distribution as the -BSSVS rate parameters. The number of parameters that achieve the point mass on 1.0 for the +BSSVS framework are Poisson distributed with a mean of 9.0 (for the +BSSVS(P) method) and uniformly distributed for the +BSSVS(U) method. For the GLM framework, there are 14 and 16 regression parameters (i.e. predictors) for the GLM(–SS) and GLM(+SS) methods, respectively, as outlined below. The regression parameters are each a priori in part a mixture of point-mass on 0 and in part normally distributed with a mean of 0 and a variance of 4.0 [12].

Sequence subsampling. In order to investigate the effects of sampling biases, we performed multiple analyses using random samples from our full set of 1,163 sequences. We created six independent sequence samples by selecting 25% of the sequences in each region at random without replacement and assume that each is representative of the entire flu season. These samples allow us to reveal whether the three frameworks will agree on the root location, root state posterior probability, height, and other trends in the phylogenies as well as show the reproducibility of the support for our GLM predictor variables. We did not identify any duplicate sequences from the same discrete state in any of the six samples. We aligned these six samples, each of which contained 285 sequences, using MAFFT v7.017 in Geneious Pro v.6.1.8 (Biomatters Ltd., Auckland, New Zealand). We treated each alignment as an independent dataset for our phylogeographic reconstructions and report all GISAID accession numbers and discrete state assignments in S1 Table. The six samples and six coalescent priors result in 180 total models, 36 from each of the -BSSVS, +BSSVS(P), +BSSVS(U), GLM(–SS), and GLM (+SS) methods.

GLM predictors

Human population and age. We obtained population estimates and land area per state from the U.S. Census Bureau (USCB) MAF/TIGER database (https://www.census.gov/). Population data are released annually and represent the population as of 2014-07-01 for the 2014-15 flu season, and we used these values to create a density per region. We also obtained the median age per state from the USCB and used these values as a separate predictor, aggregated by region.

Temperature and precipitation. For our climate predictors, we obtained data from the National Climatic Data Center of the National Oceanic and Atmospheric Administration (NOAA). We collected temperature and precipitation data for the 30-year climate normal from 1981–2010 for the 9,359 stations in the contiguous 48 states, not including the District of Columbia. As we are interested in the typical temperatures and precipitations observed during a flu season, we computed the average of all September-October-November, December-January-February, and March-April-May summary datasets from stations in each region. We take these values for temperature (in degrees Celsius) and precipitation (in centimeters) to represent the typical flu season climate for each region.

Influenza vaccination rates. We obtained state-level data on the vaccination rates for the 2014–15 flu season from FluVaxView by the Centers for Disease Control and Prevention (CDC) [26] and aggregated them to a region-wide average. These data represent all individuals at least six months of age that received the annual flu vaccine at any point in time during the season.

Air travel. In order to account for travel between the ten regions, we obtained data from the Official Airline Guide, Ltd. as the number of seats on domestic flights from each airport each other airport within the contiguous U.S. for the 2012 calendar year. We assumed that the number of seats is proportional to the number of passengers on each flight and that the 2012 travel data is proportional to that of 2014–15. We discretized the data from each individual airport into a total number per HHS region to create a matrix of travel flux. These data do not include flights originating from international locations and thus strictly represent passenger flux among the ten HHS regions used in this study. We held this predictor constant through each of the six samples.

Glycoprotein content. Influenza vaccines are designed to induce neutralizing antibodies of both the hemagglutinin and neuraminidase viral surface glycoproteins [27] in order to protect against future infections with similar antigenic properties to the vaccinated strain [28].

Predictor	Mean	SD	Median	IQR
Population Density (people/mi ²)	165.9	141.0	143.9	161.3
Median Age (years)	38.0	1.6	37.8	2.0
Vaccination Rate (%)	42.6	3.5	43.2	4.5
Temperature (°C)	7.7	4.1	6.5	6.5
Precipitation (cm)	22.4	7.0	23.7	8.2
Low Glycoprotein Content (%, overall)	88.3	3.7	87.8	3.1
Sample Size *	28.5	11.5	27.5	16
Air Travel **	6.1E+06	6.0E+06	4.1E+06	6.7E+06

Table 3. Summary statistics of the predictors used in this study for the ten discrete states.

* Accession numbers for the samples and location data are provided in <u>S1 Table</u>.

**Air travel represents the indicated statistic among all 90 pairwise region-to-region combinations.

doi:10.1371/journal.pcbi.1005389.t003

The glycoprotein (GP) content of a sampled virus thus provides an indication of the sample's similarity to the strain vaccinated against during that season. Of the 1,163 sequences in our dataset, 533 (46%) contained metadata regarding the GP content of the sample. The authors annotated these sequences with the binary "LOW GP" or "GP" to represent the similarity of the GP to the A/Texas/50/2012 (H3N2)-like virus strain vaccinated against during the 2014–15 flu season [29]. For each sample, we calculated the proportion of sequences with "LOW GP" to the total sequences with known antigenic content per region as a measure of the circulating strain's disparity from the strain vaccinated against. This is the only predictor in which the values are not fixed among the six samples.

Sample size. Previous phylogeographic studies using GLMs have included and found strong posterior support for sample size at the location of origin and/or the location of destination [12, 13] so we included both as predictors (Table 3) in the GLM(+SS) runs. The GLM (+SS) runs thus contain 16 predictors while the GLM(-SS) run contain 14 predictors.

Influenza phylogeography

Molecular clock fitting. We performed a preliminary analysis with Path-O-Gen v1.4 (http://tree.bio.ed.ac.uk/software/pathogen/) which showed that relaxed molecular clocks may have overparameterized our models. We therefore selected a strict molecular clock with a rate of 0.001 substitutions per site per year.

Coalescent priors and substitution model. In addition to the three reconstruction methods and six sequence samples, we also investigated six coalescent priors in this study: constant size [30], exponential growth [31], logistic growth [31], expansion growth [31], Bayesian Skyline [32], and Bayesian Skygrid [33]. Thus, we completed 180 individual ancestral state phylogeographic reconstructions, one for each sample/coalescent prior/reconstruction method combination (e.g. Sample 1/constant size/GLM, Sample 1/constant size/+BSSVS(P), Sample 1/constant size/-BSSVS, etc.). We specified an HKY+G [34] substitution model following recent phylogenetic studies of H3N2 [12, 35] and preliminary performance analyses using other substitution models. We enabled each of the six samples to parameterize the diffusion process between HHS regions using the -BSSVS, +BSSVS(P), +BSSVS(U), GLM(-SS), and GLM(+SS) methods. We evaluated each model using the BEAST v1.8.4 software package [36] with a chain length of 100 M, logging estimates every 10,000 steps while specifying a single seed across all models. These methods aim to minimize all sources of variance but the randomly selected sequences, tree priors, and glycoprotein content.

Analysis of support for models. We used path sampling (PS) and stepping-stone sampling (SSS) to estimate marginal likelihoods of each model, as this procedure has been shown to be an improvement over harmonic mean estimators [19, 20]. Here, we specify a chain length of 1M with 100 path steps, logging every 1,000 steps. For the GLM predictors, we obtained the mean posterior probability of inclusion, BF support values, and the contribution of each predictor to the log-linear rate matrix. In order to determine the impact of geography on the phylogeny, we utilized Bayesian Tip-association Significance Testing (BaTS) [16]. This application tests the null hypothesis that other than by chance, adjoining tips are not more likely to share the same discrete traits. Here, we used our ten HHS regions as discrete traits to be tested under this null hypothesis.

Comparison of phylogenies. We used TreeAnnotator v1.8.4 to construct a maximum clade credibility (MCC) tree for each of the 180 runs after discarding the first 10% of trees as burnin. We viewed and annotated the MCC trees using FigTree v1.4.2 for direct comparison of the ancestral state reconstructions. From each MCC tree, we recorded the root state, root height and its 95% Bayesian credible interval, root state posterior probability, and the location of all nodes with a height exceeding one year. We also calculated the Kullback-Leibler (KL) divergence at the root state of each model. Here, we assumed two different prior probabilities at each discrete state: a uniform prior probability per discrete state (i.e. 0.1 for each of the ten discrete states), and second, a prior probability that is proportional to the number of taxa from that state (e.g. as 26 of 285 taxa were sampled in Region 1 we set its prior probability to 26/ 285 = 0.0912). The latter approach allows us to account for potential sampling bias in the KL calculations. For several GLMs, we found that the posterior probability of at least one root state was zero, which yields a KL divergence of infinity. In order to present a finite KL value, we assigned these states a posterior probability of 1.0×10^{-16} and subtracted this artificial probability from the most probable root state. As an additional step to investigate possible sampling bias, we calculated the Pearson correlation coefficient (r) between the sample size for each of the ten discrete states and its corresponding root state posterior probability for each individual model.

Data availability. We have made the XML file and MCC phylogeny for each of our 180 models available for download at https://figshare.com/projects/Magee-Flu-PLoS/16638. We have also made available the six sequence alignments as well as the full set of 1,163 unaligned sequences from which we created our samples.

Supporting information

S1 Table. GISAID accession numbers and discrete states for the 285 sequences in each sample.

(CSV)

S1 Fig. Model comparisons for the 180 analyses. (A) Log marginal likelihood obtained via path sampling (PS). (B) Log marginal likelihood obtained via stepping-stone sampling (SSS). Metrics are shown for each sample, prior, and method. (TIFF)

S2 Fig. Association index scores obtained via BaTS. For each model, we show the null mean (larger value) and observed mean (smaller value) and their respective 95% confidence intervals. For each model, we observe p < 0.0001 between the null and observed means. (TIFF)

S3 Fig. Individual root state posterior probabilities and potential sampling bias analyses. (A) Root state posterior probability from the MCC tree of each model. The corresponding

root state is shown below each bar. See Fig <u>3B</u> for the locations of these root states. (B) Pearson's r correlation coefficient between the number of sequences per discrete state and the RSPP for each discrete state in each model.

(TIFF)

S4 Fig. Individual Kullback-Leibler divergence statistics of the root state prior and posterior probabilities for each model. (A) Values are calculated assuming a uniform prior probability per discrete state. (B) Values are calculated assuming a prior probability proportional to the number of sequences per discrete state. (TIFF)

S5 Fig. Root heights for the MCC phylogenies. Mean heights are represented by the colored circles with 95% Bayesian credible intervals shown as error bars. (TIFF)

S6 Fig. Posterior regression coefficients of all predictors per sample and prior for the GLM (-**SS) runs.** Predictor abbreviations are: air travel (AT), glycoprotein content (GP), median age (MA), precipitation (PC), population density (PD), sample size (SS), temperature (TP) and vaccination rate (VR), each evaluated from both region of origin (O) and region of destination (D). (TIFF)

S7 Fig. Posterior inclusion probabilities of all predictors per sample and prior for the GLM(-SS) runs. We consider predictors with inclusion probabilities exceeding the dotted horizontal line, which corresponds to BF = 3.0, to be supported in that model. Predictor abbreviations are: air travel (AT), glycoprotein content (GP), median age (MA), precipitation (PC), population density (PD), sample size (SS), temperature (TP) and vaccination rate (VR), each evaluated from both region of origin (O) and region of destination (D). (TIFF)

S8 Fig. Posterior regression coefficients of all predictors per sample and prior for the GLM (+**SS) runs.** Predictor abbreviations are: air travel (AT), glycoprotein content (GP), median age (MA), precipitation (PC), population density (PD), sample size (SS), temperature (TP) and vaccination rate (VR), each evaluated from both region of origin (O) and region of destination (D). (TIFF)

S9 Fig. Posterior inclusion probabilities of all predictors per sample and prior for the GLM(+SS) runs. We consider predictors with inclusion probabilities exceeding the dotted horizontal line, which corresponds to BF = 3.0, to be supported in that model. Predictor abbreviations are: air travel (AT), glycoprotein content (GP), median age (MA), precipitation (PC), population density (PD), sample size (SS), temperature (TP) and vaccination rate (VR), each evaluated from both region of origin (O) and region of destination (D). (TIFF)

Acknowledgments

We acknowledge the authors, originating and submitting laboratories of the sequences from GISAID's EpiFlu Database on which this research is based.

Author Contributions

Conceptualization: DM MS.

Data curation: DM MS.

Formal analysis: DM MS.

Funding acquisition: MS.

Investigation: DM MS.

Methodology: DM MS.

Project administration: MS.

Resources: MAS MS.

Software: DM MAS.

Supervision: MS.

Validation: DM MS.

Visualization: DM MS.

Writing - original draft: DM.

Writing - review & editing: MAS MS.

References

- Lemey P, Rambaut A, Drummond AJ, Suchard MA. Bayesian phylogeography finds its roots. PLoS computational biology. 2009; 5(9):e1000520. Epub 2009/09/26. doi: 10.1371/journal.pcbi.1000520 PMID: 19779555
- Kuo L, Mallick B. Variable selection for regression models. The Indian Journal of Statistics, Series B. 1998:65–81.
- Chapman H, George EI, McCulloch RE. The Practical Implementation of Bayesian Model Selection. IMS Lecture Notes—Monograph Series. 2001; 38:65–134.
- Bedford T, Riley S, Barr IG, Broor S, Chadha M, Cox NJ, et al. Global circulation patterns of seasonal influenza viruses vary with antigenic drift. Nature. 2015; 523(7559):217–20. Epub 2015/06/09. doi: 10. 1038/nature14460 PMID: 26053121
- Nelson MI, Viboud C, Vincent AL, Culhane MR, Detmer SE, Wentworth DE, et al. Global migration of influenza A viruses in swine. Nature communications. 2015; 6:6696. Epub 2015/03/31. doi: 10.1038/ ncomms7696 PMID: 25813399
- Pollett S, Nelson MI, Kasper M, Tinoco Y, Simons M, Romero C, et al. Phylogeography of Influenza A (H3N2) Virus in Peru, 2010–2012. Emerging infectious diseases. 2015; 21(8):1330–8. Epub 2015/07/ 22. doi: 10.3201/eid2108.150084 PMID: 26196599
- Scotch M, Mei C, Makonnen YJ, Pinto J, Ali A, Vegso S, et al. Phylogeography of influenza A H5N1 clade 2.2.1.1 in Egypt. BMC genomics. 2013; 14:871. Epub 2013/12/12. doi: 10.1186/1471-2164-14-871 PMID: 24325606
- Grenfell BT, Pybus OG, Gog JR, Wood JL, Daly JM, Mumford JA, et al. Unifying the epidemiological and evolutionary dynamics of pathogens. Science (New York, NY). 2004; 303(5656):327–32. Epub 2004/01/17.
- Su YC, Bahl J, Joseph U, Butt KM, Peck HA, Koay ES, et al. Phylodynamics of H1N1/2009 influenza reveals the transition from host adaptation to immune-driven selection. Nature communications. 2015; 6:7952. Epub 2015/08/08. doi: 10.1038/ncomms8952 PMID: 26245473
- Koelle K, Rasmussen DA. The effects of a deleterious mutation load on patterns of influenza A/H3N2's antigenic evolution in humans. eLife. 2015; 4:e07361. Epub 2015/09/16. doi: <u>10.7554/eLife.07361</u> PMID: 26371556
- Arafa A, El-Masry I, Kholosy S, Hassan MK, Dauphin G, Lubroth J, et al. Phylodynamics of avian influenza clade 2.2.1 H5N1 viruses in Egypt. Virology journal. 2016; 13:49. Epub 2016/03/24. doi: 10.1186/s12985-016-0477-7 PMID: 27000533
- Lemey P, Rambaut A, Bedford T, Faria N, Bielejec F, Baele G, et al. Unifying viral genetics and human transportation data to predict the global transmission dynamics of human influenza H3N2. PLoS pathogens. 2014; 10(2):e1003932. Epub 2014/03/04. doi: 10.1371/journal.ppat.1003932 PMID: 24586153

- Magee D, Beard R, Suchard MA, Lemey P, Scotch M. Combining phylogeography and spatial epidemiology to uncover predictors of H5N1 influenza A virus diffusion. Archives of virology. 2015; 160(1):215–24. Epub 2014/10/31. doi: 10.1007/s00705-014-2262-5 PMID: 25355432
- Graf T, Vrancken B, Maletich Junqueira D, de Medeiros RM, Suchard MA, Lemey P, et al. Contribution of Epidemiological Predictors in Unraveling the Phylogeographic History of HIV-1 Subtype C in Brazil. Journal of virology. 2015; 89(24):12341–8. Epub 2015/10/02. doi: <u>10.1128/JVI.01681-15</u> PMID: 26423943
- Yang W, Lipsitch M, Shaman J. Inference of seasonal and pandemic influenza transmission dynamics. Proceedings of the National Academy of Sciences of the United States of America. 2015; 112(9):2723– 8. Epub 2015/03/03. doi: 10.1073/pnas.1415012112 PMID: 25730851
- Parker J, Rambaut A, Pybus OG. Correlating viral phenotypes with phylogeny: accounting for phylogenetic uncertainty. Infection, genetics and evolution: journal of molecular epidemiology and evolutionary genetics in infectious diseases. 2008; 8(3):239–46. Epub 2007/10/09. doi: 10.1016/j.meegid.2007.08. 001 PMID: 17921073
- Rambaut A, Pybus OG, Nelson MI, Viboud C, Taubenberger JK, Holmes EC. The genomic and epidemiological dynamics of human influenza A virus. Nature. 2008; 453(7195):615–9. Epub 2008/04/18. doi: 10.1038/nature06945 PMID: 18418375
- Viboud C, Alonso WJ, Simonsen L. Influenza in Tropical Regions. PLoS Med. 2006; 3(4):e89. doi: 10. 1371/journal.pmed.0030089 PMID: 16509764
- Baele G, Lemey P, Bedford T, Rambaut A, Suchard MA, Alekseyenko AV. Improving the accuracy of demographic and molecular clock model comparison while accommodating phylogenetic uncertainty. Molecular biology and evolution. 2012; 29(9):2157–67. Epub 2012/03/10. doi: 10.1093/molbev/mss084 PMID: 22403239
- Baele G, Li WL, Drummond AJ, Suchard MA, Lemey P. Accurate model selection of relaxed molecular clocks in bayesian phylogenetics. Molecular biology and evolution. 2013; 30(2):239–43. Epub 2012/10/ 24. doi: 10.1093/molbev/mss243 PMID: 23090976
- Viboud C, Bjornstad ON, Smith DL, Simonsen L, Miller MA, Grenfell BT. Synchrony, waves, and spatial hierarchies in the spread of influenza. Science (New York, NY). 2006; 312(5772):447–51. Epub 2006/ 04/01.
- Bahl J, Nelson MI, Chan KH, Chen R, Vijaykrishna D, Halpin RA, et al. Temporally structured metapopulation dynamics and persistence of influenza A H3N2 virus in humans. Proceedings of the National Academy of Sciences of the United States of America. 2011; 108(48):19359–64. Epub 2011/11/16. doi: 10.1073/pnas.1109314108 PMID: 22084096
- Frost SD, Pybus OG, Gog JR, Viboud C, Bonhoeffer S, Bedford T. Eight challenges in phylodynamic inference. Epidemics. 2015; 10:88–92. Epub 2015/04/07. doi: 10.1016/j.epidem.2014.09.001 PMID: 25843391
- Baele G, Suchard MA, Rambaut A, Lemey P. Emerging Concepts of Data Integration in Pathogen Phylodynamics. Systematic biology. 2016. Epub 2016/06/12.
- HHS. Regional Offices 2014 [updated Apr 15; cited 2016 Jul 5]. http://www.hhs.gov/about/agencies/ iea/regional-offices/index.html.
- CDC. 2014–15 Flu Season 2015 [updated Sep 17; cited 2015 Feb 2]. http://www.cdc.gov/flu/ fluvaxview/1415season.htm.
- Cobbin JC, Verity EE, Gilbertson BP, Rockman SP, Brown LE. The source of the PB1 gene in influenza vaccine reassortants selectively alters the hemagglutinin content of the resulting seed virus. Journal of virology. 2013; 87(10):5577–85. Epub 2013/03/08. doi: 10.1128/JVI.02856-12 PMID: 23468502
- 28. Couch RB, Kasel JA. Immunity to influenza in man. Annual review of microbiology. 1983; 37:529–49. Epub 1983/01/01. doi: 10.1146/annurev.mi.37.100183.002525 PMID: 6357060
- CDC. What You Should Know for the 2014–2015 Influenza Season 2016 [updated 18 Feb 2016; cited 2016 1 Jun]. http://www.cdc.gov/flu/pastseasons/1415season.htm.
- 30. Kingman J. The coalescent. Stochastic Processes and their Applications. 1982; 13(3):235–48.
- Griffiths RC, Tavare S. Sampling theory for neutral alleles in a varying environment. Philosophical transactions of the Royal Society of London Series B, Biological sciences. 1994; 344(1310):403–10. Epub 1994/06/29. doi: 10.1098/rstb.1994.0079 PMID: 7800710
- **32.** Drummond AJ, Rambaut A, Shapiro B, Pybus OG. Bayesian coalescent inference of past population dynamics from molecular sequences. Molecular biology and evolution. 2005; 22(5):1185–92. Epub 2005/02/11. doi: 10.1093/molbev/msi103 PMID: 15703244
- Gill MS, Lemey P, Faria NR, Rambaut A, Shapiro B, Suchard MA. Improving Bayesian population dynamics inference: a coalescent-based model for multiple loci. Molecular biology and evolution. 2013; 30(3):713–24. Epub 2012/11/28. doi: 10.1093/molbev/mss265 PMID: 23180580

- Hasegawa M, Kishino H, Yano T. Dating of the human-ape splitting by a molecular clock of mitochondrial DNA. Journal of molecular evolution. 1985; 22(2):160–74. Epub 1985/01/01. PMID: 3934395
- **35.** Horm SV, Mardy S, Rith S, Ly S, Heng S, Vong S, et al. Epidemiological and virological characteristics of influenza viruses circulating in Cambodia from 2009 to 2011. PloS one. 2014; 9(10):e110713. Epub 2014/10/24. doi: 10.1371/journal.pone.0110713 PMID: 25340711
- 36. Drummond AJ, Suchard MA, Xie D, Rambaut A. Bayesian phylogenetics with BEAUti and the BEAST 1.7. Molecular biology and evolution. 2012; 29(8):1969–73. Epub 2012/03/01. doi: 10.1093/molbev/ mss075 PMID: 22367748